\begin{table}%t3
\caption{\label{summary_tab}Determination of main HFI parameters status and references for their values.}
\par
%\centering
\par
\begin{tabular}{lcc}
      \hline
      \hline \noalign{\smallskip}
      &Status and/or determination error&Reference\\
      \hline
      Beams and far side lobes&computed from front horns and telescope 
measurements&Maffei et~al. (\cite{maffei1}), Tauber et~al. (\cite{tauber2})\\
      \hline
      Spectral bands&0.1~cm$^{-1}$ resolution, $\nu<400$~GHz: 3\% error, 
$\nu>400$~GHz: 1\% error&Ade et~al. (\cite{ade1}) \\
      &\emph{final determination within requirement}&\\
      \hline
      Polarisation orientation&0.3$^{\circ}$/0.6$^{\circ}$/2.1$^{\circ}$ (min/avg/max) 
for Polarisation Sensistive Bolometers&Rosset et~al. (\cite{rosset1}) \\
      Cross-polarisation leakage&0.1\%/0.2\%/2.2\% (min/avg/max) for PSB&\\
      &\emph{final determination requires sky data}&\\
      \hline
      4~K stage emissivity&better than 1\%&Sect.~\ref{emissivity4K1.6K}\\
      1.6~K stage emissivity&better than 3\% for $\nu<300$~GHz&Sect.~\ref{emissivity4K1.6K}\\
      &\emph{within requirements to allow correction of remaining}&\\
      &\emph{systematics coming from thermal stability below}&\\
      \hline
      Thermal stability&determined within range [1~mHz, 100~Hz]&Sect.~\ref{thermalstab}\\
      &\emph{flight data required for the final stability determination}&\\
      \hline
      Linearity&0.1\% determination within requirements&Sect.~\ref{linear}\\
      \hline
      Time response&measured to better than 0.1\% for 20 detectors&Sect.~\ref{timeresp}\\
      &\emph{sky data expected for the complete set of detectors}&\\
      \hline
      Response&3\% determination error& Sects.~\ref{opteff},~\ref{linear}\\
      &\emph{sky data required to provide the absolute calibration}&\\
      \hline
      Total detector noise&explored within range [1~mHz, 100~Hz]&Sect.~\ref{noisean}, 
Lamarre et~al. (\cite{lamarre1}) \\
      Compatibility&checked to 1\% to 3\% of noise level&Sect.~\ref{labcsl}\\
      &\emph{flight data required to provide the ultimate values,}&\\
      &\emph{including particle hits rate}&\\
      \hline
      Optical crosstalk&down to 60~dB&Sect.~\ref{ocrosstalk}\\
      Electrical crosstalk&down to 80~dB&Sect.~\ref{ecrosstalk}\\
      &\emph{determination within requirement, sky data to reach goal of 80~dB}&\\
      \hline
      \end{tabular}
\end{table}